Author's response to reviews

Title: Genome instability in blood cells of a BRCA1+ breast cancer family

Authors:

Fengxia Xiao (fengxia.xiao@unmc.edu)
Yeong C Kim (yeong.kim@unmc.edu)
Carrie Snyder (csnyder@creighton.edu)
Hongxiu Wen (wenhongxiu@gmail.com)
Pei Xian Chen (peixian.chen@unmc.edu)
Jiangtao Luo (jiangtao.luo@unmc.edu)
Dina Becirovic (DinaBecirovic@creighton.edu)
Bradley Downs (bradley.downs@unmc.edu)
Kenneth H Cowan (kcowan@unmc.edu)
Henry Lynch (htlynch@creighton.edu)
San Ming Wang (sanming.wang@unmc.edu)

Version: 7
Date: 29 April 2014

Author's response to reviews:

the acknowledgement section has been moved after the authors contribution.

the Availability of supporting data section has been added before Competing interests:

The exome data generated by the study were deposited in NCBI (Accession number SRR949927). url: